Author's response to reviews

Title: Atrial Myxoma Presenting with Orthostatic Hypotension: a case report

Authors:

Ralph Vicari (ralph.vicari@mima.com)
Enrique Polanco (enrique.polanco@mima.com)
Norberto Schechtmann (norberto.schechtmann@mima.com)
Jose Santiago (jose.santiago@mima.com)
Kautilya Shaurya (k.shaurya@umiami.edu)
Michael Halstead (mhalstea@tulane.edu)
Danielle Marszal (danielle.marszal@hotmail.com)
Tamsin Grosskreutz (tgrosskr@fau.edu)
Shalini Thareja (sthareja@columbia.edu)

Version: 4 Date: 5 January 2009

Author's response to reviews:

The mechanism believed to cause the orthostasis presented is now clearly stated in the Case Report.

The reference numbering now correlates to the article properly.

Other causes of orthostatic hypotension are now stated in the discussion and it is concluded that after they are ruled out, a two-dimensional echocardiogram may provide detection of intracardiac growths.

Overall, the English of the manuscript was improved.